Author’s response to reviews

Title: Chemical cystitis due to crystal violet: A case report

Authors:

Takashi Hashimoto (ha-tkc@tokyo-med.ac.jp)
Makoto Ohori (mak_ohori@yahoo.co.jp)
Takeshi Kashima (takeshikasima922@yahoo.co.jp)
Hidenobu Yamamoto (customroadbike610@yahoo.co.jp)
Masaaki Tachbana (tachi@tokyo-med.ac.jp)

Version: 2 Date: 13 December 2012

Author's response to reviews: see over
On behalf of all the authors, I would like to ask you to consider our manuscript entitled "Chemical cystitis due to crystal violet" for publication in Journal of Medical Case Reports as a case report. We experienced a case of chemical cystitis due to intravesical instillation of crystal violet. A gynecologist instilled crystal violet into the bladder to confirm the presence of vesicovaginal fistula. The case which caused the membrane obstacle by the direct toxicity of a medicine is very rare. Related with crystal violet, there had been only 3 case reports published before. Although this case had trouble for medical treatment very much, it has improved over many hours. We treated the patient with conservative treatment using non-steroid anti-inflammatory drug and glucocorticoid.

We feel that findings from this case will be of special interest to the readers of Journal of Medical Case Reports. And we must call attention to some gynecologist who uses crystal violet.

This manuscript has not been published and is not under consideration for publication elsewhere. All the authors have read the manuscript and have approved this submission. The authors report no conflicts of interest.

Sincerely,

Takeshi Hashimoto, MD
Department of Urology,
Tokyo Medical University,
6-7-1, Nishishinjuku, Shinjuku-ku, Tokyo,
160-0023,
Japan
Tel: +81-3-3342-6111
Fax: +81-3-3344-4813
E-mail address: ha-tkc@tokyo-med.ac.jp